Author's response to reviews

Title: Zuo Jin Wan reverses P-gp-mediated multidrug resistance by inhibiting activation of the PI3K/Akt/NF-kappaB pathway

Authors:

hua sui (syh0808@163.com)
shufang pan (panshufang2010@126.com)
bao hui jin (jbhrachel@gmail.com)
xuan liu (lxtw@hotmail.com)
li hong zhou (zlhtcm@hotmail.com)
feng gang hou (fghou555@hotmail.com)
hui rong zhu (zhu_huirong@126.com)
lian jin ren (renjianlin666@126.com)
qi li (lzwf@hotmail.com)
wen hai wang (wwhdoctor@gmail.com)
xia o ling fu (fuxiaoling2014@163.com)

Version: 5 Date: 8 July 2014

Author's response to reviews: see over
Dear Editor,

We have carefully studied the comments of the editor, performed a author Email list in title and add the Figure Legends as suggested accordingly. At beginning, the Figure Legends has been in the manuscript with the Figures. However, it has been removed with the Figure when we were asked it needed not to put the figures in the manuscript.

Best,

Hua Sui(first author)